Author's response to reviews

Title: Insulin treatment corrects hepcidin but not YKL-40 levels in persons with type 2 diabetes mellitus matched by Body Mass Index, Waist-to-Height Ratio, C-Reactive Protein and Creatinine

Authors:

Driton Vela (dritonvela@hotmail.com)

Jovica Leshoski (jovicalesoski@yahoo.com)

Zana Vela (zanavela78@gmail.com)

Muharrem Jakupaj (muharrem.jakupi@uni-pr.edu)

Mitko Mladenov (m.mitko@gmail.com)

Ramadan Sopi (ramadan.sopi@uni-pr.edu)

Version: 3 Date: 17 Aug 2017

Author’s response to reviews:

Dear Editor,

We have made the following changes in response to Editor Comments:

Editor Comments:

Please revise the ethics heading under your declarations section, to provide full details of ethical approval and consent to participate.

Response to Editor Comments: In response to Editor comment regarding the ethics heading, we have revised the heading as requested.

Kind regards,

Ramadan B. Sopi